Author's response to reviews

Title: A quasi-experimental test of an intervention to increase the use of thiazide-based treatment regimens for people with hypertension

Authors:

Carol M Ashton MD MPH (cashton@uab.edu)
Myrna M Khan PhD MBA (mkhan2@swbell.net)
Michael L Johnson PhD (mike.johnson@uh.edu)
Annette Walder MS (annette.walder@med.va.gov)
Elizabeth Stanberry PharmD (elizabeth.stanberry@med.va.gov)
Rebecca J Beyth MD MS (rbeyth@aging.ufl.edu)
Tracie C Collins MD MPH (tcc@umn.edu)
Howard S Gordon MD (hsg@uic.edu)
Paul Haidet MD MPH (phaidet@bcm.tmc.edu)
Barbara Kimmel MS MSc (barbara.kimmel@med.va.gov)
Anna Kolpakchi MD (kolpakchi.annal@med.va.gov)
Lee B Lu MD (lblu@bcm.tmc.edu)
Aanand D Naik MD MPH (anaik@bcm.tmc.edu)
Laura A Petersen MD MPH (laurap@bcm.tmc.edu)
Hardeep Singh MD MPH (hardeeps@bcm.tmc.edu)
Nelda P Wray MD MPH (nwray@uab.edu)

Version: 4 Date: 31 January 2007

Author's response to reviews:

In his email of 1.30.07, Dr Mittman told me no cover letter was needed with this version because it is a technical revision (copy edits) rather than a substantive revision.